Author’s response to reviews

Title: Comparison of Screening Strategies for Prevalent Vertebral Fractures in South Korea: Vertebral Fracture Assessment vs. Spine Radiography

Authors:

Sung-Hee Oh (crane999@hanmail.net)
Dam Kim (damii00@naver.com)
Young Eun Lee (o0eun0o@naver.com)
Deog-Yoon Kim (deogyoon@daum.net)
Yu Kyung Lee (yklee82@gmail.com)
Joo-Hyun Lee (jhl000@naver.com)
Sang-Cheol Bae (scbae@hanyang.ac.kr)
Yun Young Choi (yychoi@hanyang.ac.kr)
Junhee Pyo (stdpjh@gmail.com)
Jeonghoon Ahn (jahn0203@gmail.com)
Yoon-Kyoung Sung (sungyk@hanyang.ac.kr)

Version: 2 Date: 26 Jan 2018

Author’s response to reviews:

Dear Dr. Sondos Majeed,

The other authors and I greatly appreciate the reviewer’s helpful comment. We are submitting this paper as a “revised version”. We hope that this revised version of our paper will be considered suitable for publication in BMC Musculoskeletal Disorders.

With best regards,

Yoon-Kyoung Sung, MD, PhD, MPH
Professor

Department of Rheumatology, Hanyang University Hospital for Rheumatic Diseases, Seoul 04763, Republic of Korea

E-mail: sungyk@hanyang.ac.kr